Check for updates

OPEN ACCESS

EDITED BY Shangke Huang, Southwest Medical University, China

REVIEWED BY Xuejun Guo, Shanghai Jiao Tong University, China Chengwu Liu, Sichuan University, China

*CORRESPONDENCE Guibao Xiao, ⊠ 1250374226@qq.com Yanchun Yang, ⊠ 18190287267@163.com

RECEIVED 09 April 2025 ACCEPTED 19 May 2025 PUBLISHED 05 June 2025

CITATION

Lin Z, Wu L, Yu Y, Jiang J, Yang Y and Xiao G (2025) Case Report: Osimertinib-induced acute interstitial lung disease. *Front. Pharmacol.* 16:1608733. doi: 10.3389/fphar.2025.1608733

COPYRIGHT

© 2025 Lin, Wu, Yu, Jiang, Yang and Xiao. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Case Report: Osimertinib-induced acute interstitial lung disease

Zhiwu Lin¹, Liang Wu¹, Yang Yu¹, Jiudong Jiang¹, Yanchun Yang^{1*} and Guibao Xiao^{2*}

¹Department of Thoracic Surgery, Ziyang Central Hospital, Ziyang, China, ²Department of Infectious Diseases, Ziyang Central Hospital, Ziyang, China

Osimertinib is a third-generation irreversible epidermal growth factor receptor tyrosine kinase inhibitor (EGFR-TKI) that selectively targets EGFR-TKI-sensitive mutations, thereby inhibiting tumor cell proliferation, migration, and invasion. Herein, we present a case of osimertinib-induced interstitial lung disease (ILD) in an 80-year-old woman with EGFR-mutated lung adenocarcinoma. The patient was treated with osimertinib as first-line therapy for metastatic non-small cell lung cancer (NSCLC). On day 45 of treatment, she experienced acute onset of severe dyspnea, which rapidly progressed to diffuse bilateral pulmonary consolidation and profound hypoxemia. Despite discontinuation of osimertinib and administration of aggressive supportive care, her clinical condition continued to deteriorate, ultimately resulting in a fatal outcome. This case underscores the importance of monitoring respiratory symptoms in patients receiving EGFR-TKIs, promptly diagnosing ILD, and implementing early intervention to mitigate adverse outcomes.

KEYWORDS

adverse drug reaction, osimertinib, interstitial lung disease, lung cancer, EGFR-TKI

1 Introduction

Lung cancer remains the leading cause of cancer-related mortality worldwide (Siegel et al., 2014). Non-small cell lung cancer (NSCLC) constitutes approximately 80%–85% of all lung cancer cases, with lung adenocarcinoma being the most prevalent histological subtype (Wang et al., 2021). A substantial proportion of lung adenocarcinomas harbor activating mutations in the epidermal growth factor receptor (EGFR). Osimertinib, a third-generation EGFR tyrosine kinase inhibitor (TKI), has become a standard first-line treatment for patients with metastatic NSCLC harboring EGFR exon 19 deletions or exon 21 L858R mutations (Isobe et al., 2021). While gastrointestinal symptoms and skin mucosal lesions are the most frequently reported adverse effects, osimertinib-induced ILD is extremely rare yet represents a severe and potentially fatal adverse reaction (Xie et al., 2020). Herein, we report a case of an EGFR-mutated lung adenocarcinoma patient who developed refractory dyspnea and hypoxemia after 1.5 months of osimertinib therapy, ultimately leading to treatment failure and death.

2 Case description

An 80-year-old female patient was admitted to the hospital on 24 September 2024, due to a persistent cough. Chest computed tomography (CT) revealed a mass in the right upper lobe with



associated obstructive pneumonia (Figure 1A). On October 1, under local anesthesia, a needle biopsy of the lesion was performed, and the pathology confirmed adenocarcinoma. During her evaluation, an enhanced brain magnetic resonance imaging (MRI) scan revealed multiple intracranial metastatic lesions. The patient was subsequently diagnosed with right lung adenocarcinoma with intracranial metastasis (cT2aNxM1c). Gene sequencing identified an EGFR exon 19 deletion mutation. Her medical history included hypertension and cerebral atrophy; however, chronic obstructive pulmonary disease (COPD) or other pulmonary conditions had not been previously documented. Pulmonary function testing upon admission indicated an FEV1/FVC ratio of 86%. Following the diagnosis, she initiated treatment with osimertinib at a dose of 80 mg once daily on October 9.

On day 32 of treatment, the patient developed diarrhea. In response, temporary antidiarrheal medication was administered, leading to improvement in her symptoms. Subsequently, a chest computed tomography was performed, which revealed no abnormalities in the lungs (Figure 1B). The clinical response was assessed as partial remission (PR), and osimertinib treatment was continued.

On the 37th day of treatment, the patient experienced mild shortness of breath after exercise, accompanied by cough and sputum production, primarily characterized by white mucus sputum, and the symptoms were mild.

On the 45th day of treatment, the patient developed resting dyspnea. A physical examination showed an oxygen saturation level of 78% on room air, which increased to 90% with mask oxygen supplementation, accompanied by bilateral lung crackles. Laboratory findings revealed abnormal blood gas values (pH 7.46, PO2 48 mmHg, PCO2 34.7 mmHg) and negative results for respiratory pathogens. However, inflammatory markers, including C-reactive protein, interleukin levels, and neutrophil percentage (NEUT%), were markedly elevated beyond the normal range. Notably, B-type natriuretic peptide (BNP) and D-dimer levels remained within the normal range. Comprehensive assessments of cardiac, hepatic, and renal functions revealed no significant abnormalities. A diagnosis of pulmonary infection complicated by type I respiratory failure was established. Despite receiving oxygen therapy, anti-infective treatment, and hormonemediated anti-inflammatory therapy, the patient's clinical condition showed no significant improvement.

On day 48 of treatment, the patient experienced a progressive worsening of dyspnea, necessitating mechanical ventilation. Despite an oxygen concentration exceeding 60%, the arterial partial pressure of oxygen (PaO2) remained at only 40–50 mmHg. Emergency chest computed tomography (CT) revealed diffuse exudative changes in the bilateral lung parenchyma, with involvement exceeding 90% of the lung fields (Figures 2 A, B). Osimertinib was discontinued, and sulprostone was initiated for anti-infective treatment. Additionally, methylprednisolone pulse therapy (500 mg/day) was administered for 3 days, followed by maintenance therapy at 2 mg/kg/day. During this period, the patient developed bilateral spontaneous pneumothorax, which was managed with bilateral thoracic closed drainage.

On the 53rd and 57th days of treatment, C-reactive protein, interleukin, and neutrophil percentage all returned to normal levels. However, clinical symptoms remained unresolved, lung lesions showed no significant improvement, and persistent hypoxemia failed to be corrected. The patient eventually succumbed to respiratory failure on 10 December 2024. Figure 3 presented the patient's treatment timeline.

3 Discussion

Osimertinib is an oral, irreversible, third-generation EGFR tyrosine kinase inhibitor (EGFR-TKI) that exhibits significantly superior efficacy compared with first- and second-generation EGFR-TKIs (Soria et al., 2018). It is currently the standard firstline treatment for advanced non-small cell lung cancer (NSCLC) harboring EGFR mutations. Common adverse effects of osimertinib include diarrhea, rash, and oral mucosal lesions, which are generally well tolerated. However, interstitial lung disease (ILD) represents a rare but potentially fatal pulmonary complication in some patients receiving osimertinib (Fan et al., 2019). ILD encompasses a group of diffuse lung diseases affecting the lung interstitium, alveolar spaces, or bronchioles, often characterized by severe lung function impairment and hypoxemia. Targeted drug-induced ILD (TDILD) is a specific form of ILD caused by targeted therapies (Ma and Dai, 2022). According to the CTCAE classification, TDILD is categorized into four grades (National Institutes of Health, 2025),





and its histopathological features typically demonstrate diffuse alveolar damage, which is the primary cause of respiratory failure. The onset pattern of TDILD varies, manifesting as either acute or subacute onset with prominent clinical symptoms that may become life-threatening within a short period, or chronic insidious onset progressing gradually to respiratory failure. The reported incidence of osimertinib-induced interstitial pneumonia is approximately 1%-3%, with a mortality rate of 0.13% (De et al., 2019; Priantti et al., 2024). Compared with phase III clinical trial results for other EGFR-TKIs such as erlotinib and afatinib, osimertinib-associated ILD has a higher mortality rate of approximately 0.4% (Mok et al., 2017). Clinical case reports indicate that TDILD often manifests early during treatment, with a median onset time of 34 days (Zhou et al., 2024). In this case, the patient developed CTCAE Grade 4 interstitial pneumonia 44 days after initiating osimertinib, consistent with the median onset time of 2.7 months noted in the osimertinib mesylate label. The exact pathogenesis of TDILD remains unclear but may involve the role of EGFR in the repair and regeneration of alveolar epithelial cells. Some studies suggest that EGFR-TKIs might interfere with the

proliferation and differentiation of type II alveolar epithelial cells (AT2) and inhibit epithelial cell repair following injury by targeting the EGFR signaling pathway in tumor tissues, leading to alveolar structural disruption and fibrosis (Ohmori et al., 2021). In addition, EGFR-TKI can induce chronic inflammation in alveolar and bronchial epithelia, promote the migration and proliferation of fibroblasts, and stimulate the production of extracellular matrix, ultimately resulting in pulmonary fibrosis. Furthermore, these agents may also trigger immune-mediated allergic reactions, contributing to the development of ILD (Matsuno, 2012).

Pathological testing remains a reliable method for confirming TDILD; however, given the challenges in obtaining pathological tissue samples, the diagnosis of TDILD frequently depends on clinical medication history, imaging findings, and the exclusion of other potential causes (Saito and Gemma, 2012; He and Zhou, 2019). In this case, the patient's condition was critical, and the family declined needle biopsy, rendering pathological evidence unattainable. Nonetheless, based on established clinical diagnostic criteria, the patient demonstrated imaging features consistent with acute interstitial pneumonia, with involvement exceeding 75% of

lung parenchyma. Tests for relevant pathogens returned negative results, and cardiogenic causes as well as pulmonary embolism were excluded. Considering the patient's history of osimertinib treatment, a diagnosis of TDILD (CTCAE grade 4) was confirmed. Currently, the management of TDILD lacks standardization but generally involves discontinuing targeted therapy and suppressing inflammatory responses to control pulmonary interstitial fibrosis. For patients with CTCAE grade 2-4 pulmonary toxicity, it is recommended to discontinue EGFR-TKIs and initiate oxygen therapy and hormone therapy. Although there is no definitive consensus regarding the optimal dose of hormone therapy, the most commonly used regimen is prednisone at 0.5-2 mg/kg/day. For patients with CTCAE grade 4 pulmonary toxicity, methylprednisolone pulse therapy (500-1,000 mg/day) can be administered for up to 3 days to improve the condition (Chinese Anti-Cancer Association Chinese Society of Lung Cancer, 2019). TDILD progresses rapidly, and the mortality rate associated with CTCAE grade 4 is high. Early intervention is crucial for achieving better therapeutic outcomes.

4 Conclusion

Herein, we report a series of interstitial lung diseases potentially induced by osimertinib. The underlying mechanisms remain unclear but may involve multiple factors, such as inhibition of the EGFR pathway in normal lung tissue, impaired regeneration and repair processes, and abnormal immune activation. The incidence of ILD caused by osimertinib is significantly lower than that of the skin and digestive tract adverse reactions, but the mortality rate is still high due to the rapid development of the disease and timely discontinuation of the drug. Clinicians should be aware that any new respiratory symptoms in patients using EGFR-TKI should be carefully evaluated and alert for the development of ILD. Early identification and hormone treatment are essential to improve the safety of medication.

Data availability statement

The original contributions presented in the study are included in the article/supplementary material, further inquiries can be directed to the corresponding authors.

References

Chinese Anti-Cancer Association Chinese Society of Lung Cancer (2019). EGFR-TKI ADR management Chinese expert consensus. *Chin. J. lung cancer* 22 (2), 57–81. doi:10. 3779/j.issn.1009-3419.2019.02.01

De, M. F., Wu, Y. L., Jr, D. C. G., Chang, G. C., Chen, Y. M., Cho, B. C., et al. (2019). ASTRIS: a global real-world study of osimertinib in >3000 patients with EGFR T790M positive non-small-cell lung cancer. *Future Oncol.* 15, 3003–3014. doi:10.2217/fon-2019-0324

Fan, M., Mo, T., Shen, L., and Yang, L. (2019). Osimertinib-induced severe interstitial lung disease: a case report. *Thorac. cancer* 10 (7), 1657–1660. doi:10.1111/1759-7714.13127

He, Y., and Zhou, C. (2019). Tyrosine kinase inhibitors interstitial pneumonitis: diagnosis and management. *Transl. Lung Cancer Res.* 8 (S3), S318-S320–S320. doi:10.21037/tlcr.2019.05.02

Isobe, K., Yoshizawa, T., Sekiya, M., Miyoshi, S., Nakamura, Y., Urabe, N., et al. (2021). Quantification of BIM mRNA in circulating tumor cells of osimertinib-treated patients with EGFR mutation-positive lung cancer. *Respir. Investig.* 59 (4), 535–544. doi:10.1016/j.resinv.2021.03.010

Ethics statement

Written informed consent was obtained from the individual(s) for the publication of any potentially identifiable images or data included in this article.

Author contributions

ZL: Writing – review and editing. LW: Writing – original draft. YgY: Writing – original draft. JJ: Writing – review and editing, Writing – original draft. YcY: Writing – review and editing. GX: Writing – original draft, Writing – review and editing.

Funding

The author(s) declare that no financial support was received for the research and/or publication of this article.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Generative AI statement

The author(s) declare that no Generative AI was used in the creation of this manuscript.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Ma, F., and Dai, H. P. (2022). Expert consensus on diagnosis and treatment of antitumor drug-related interstitial lung disease. *Chin. J Cancer* 8 (7), 693–702. doi:10. 3760/cma.j.cn112152-20220412-00244

Matsuno, O. (2012). Drug-induced interstitial lung disease: mechanisms and best diagnostic approaches. *Respir. Res.* 13 (1), 39. doi:10.1186/1465-9921-13-39

Mok, T. S., Wu, Y. L., Ahn, M. J., Garassino, M. C., Kim, H. R., Ramalingam, S. S., et al. (2017). Osimertinib or platinum-pemetrexed in EGFR t790m-positive lung cancer. *N. Engl. J. Med.* 376 (7), 629–640. doi:10.1056/NEJMoa1612674

National Institutes of Health (2025). Common Terminolog y Cr iter ia for Adverse Events (CTCAE)v5.0. Available online at: https://ctep.cancer.gov/protocoldevelopment/electronic_applications/ctc.htm#ctc_50.

Ohmori, T., Yamaoka, T., Ando, K., Kusumoto, S., Kishino, Y., Manabe, R., et al. (2021). Molecular and clinical features of EGFR-TKI-associated lung injury. *Int. J. Mol. Sci.* 22 (2), 792. doi:10.3390/ijms22020792

Priantti, J. N., Vilbert, M., Moraes, F. C. A., Michelon, I., Castro, C., Leighl, N. B., et al. (2024). Safety and efficacy of osimertinib in patients with NSCLC and uncommon tumoralEGFRmutations: a systematic review and meta-analysis. *J. Clin. Oncol.* 42 (16_Suppl. 1), 8642. doi:10.1200/JCO.2024.42.16_suppl.8642

Saito, Y., and Gemma, A. (2012). Current status of DILD in molecular targeted therapies. Int. J. Clin. Oncol. 17 (6), 534–541. doi:10.1007/s10147-012-0494-5

Siegel, R., Ma, J., Zou, Z., and Jemal, A. (2014). Cancer statistics, 2014. *Can- cer J. Clin.* 64 (1), 9–29. doi:10.3322/caac.21208

Soria, J. C., Ohe, Y., Vansteenkiste, J., Reungwetwattana, T., Chewaskulyong, B., Lee, K. H., et al. (2018). Osimertinib in untreated EGFR-mutated advanced non-small-cell lung cancer. *N. Engl. J. Med.* 378 (2), 113–125. doi:10.1056/NEJMoa1713137

Wang, Y., Zhou, Z., Chen, L., Li, Y., and Chu, X. (2021). Identification of key genes and biological pathways in lung adenocarcinoma via bioinformatics analysis. *Mol. Cell. Biochem.* 476 (2), 931–939. doi:10.1007/s11010-020-03959-5

Xie, X., Wang, X., Wu, S., Yang, H., Liu, J., Chen, H., et al. (2020). Fatal toxic effects related to EGFR tyrosine kinase inhibitors based on 53 cohorts with 9,569 participants. *J. Thorac. Dis.* 12 (8), 4057–4069. doi:10.21037/JTD-19-4000A

Zhou, S., Kishi, N., Alerasool, P., and Rohs, N. C. (2024). Adverse event profile of epidermal growth factor receptor tyrosine kinase inhibitors for non-small cell lung cancer: an updated meta-analysis. *Target. Oncol.* 19 (4), 547–564. doi:10.1007/s11523-024-01073-w